Author's response to reviews

Title: Functional characterization of the complement receptor type 1 and its circulating ligands in patients with schizophrenia

Authors:

Arsen Arakelyan (aaraekelyan@sci.am)
Roksana Zakharyan (r_zakharyan@mb.sci.am)
Aren Khoyetsyan (akhoyetsyan@mb.sci.am)
David Poghosyan (d_poghosyan@mb.sci.am)
Rouben Aroutiounian (genetik@ysu.am)
Frantisek Mrazek (frantisek.mrazek@fnol.cz)
Martin Petrek (martin.petrek@fnol.cz)
Anna Boyajyan (aboyajyan@sci.am)

Version: 4 Date: 24 August 2011

Author's response to reviews: see over
Dear Dr Norton,

We are submitting out finally revised manuscript entitled “Functional characterization of the complement receptor type 1 and its circulating ligands in patients with schizophrenia” by Arakelyan et al. for publication in BMC Clinical Pathology.

We made all changes recommended for the previous version of the manuscript submitted to BMC Clinical Pathology on August 22, 2011. In particular:

**Major revisions:**
All references are cited in numerical order now.

**Minor revisions:**
We found no typographical errors to be changed.

All the authors declare,
1) that the manuscript has been seen and approved by all authors, it is not under active consideration for publication,
2) that they provided the consent to publication in BMC Clinical Pathology,
3) that the manuscript has not been accepted for publication, nor has it been published, in full or in part (except in abstract form),
4) that the manuscript has not been published elsewhere and, if accepted in BMC Clinical Pathology, will not be republished in any other journal in the same or similar form,
5) no financial or other relationship that might lead to a conflict of interest.

Also, we declare that it adheres to the editorial policies of BMC Clinical Pathology detailed in the instruction for authors.

We would like to thank You for the evaluation of our manuscript.

Yours sincerely,

Prof. Anna Boyajyan

Corresponding author
Institute of Molecular Biology NAS RA
Address: 7 Hasratyan St., 0014, Yerevan, Armenia
Telephone: +37410(281626)
Fax: +37410(282061)
E-mail: aboyajyan@sci.am
Web: www.molbiol.sci.am